Author's response to reviews

Title: Arthrogryposis in infancy, multidisciplinary approach. Case report.

Authors:

Anna Binkiewicz-Glinska (abinkiewicz@gumed.edu.pl)
Agnieszka Sobierajska-Rek (incertus@wp.pl)
Stanislaw Bakula (rehabil@gumed.edu.pl)
Jolanta Wierzba (jolantawierzba@wp.pl)
Konrad Drewek (kdrewek@gumed.edu.pl)
Ireneusz M Kowalski (i.kowalski@wp.pl)
Katarzyna Zaborowska-Sapeta (katezab@poczta.onet.pl)

Version: 9 Date: 25 September 2013

Author's response to reviews: see over
We would like to re-submit our manuscript entitled “Arthrogryposis in infancy, multidisciplinary approach. Case report.” for You to consider for publication in BMC Pediatrics. We have carefully analysed the reviewer’s and editor’s comments and tried our best to improve our manuscript as to meet the standards of the journal.

The main shortcoming was the quality of grammar/spelling and abbreviations. The manuscript has been proofread by a fluent English speaking professional again and all the concerns pointed out in your comments have been carefully checked and corrected.

Yours faithfully,

Anna Binkiewicz-Glinska